Author's response to reviews

**Title:** The Expert Network and Electronic Portal for children with respiratory and allergic symptoms: rationale and design.

**Authors:**

Kim Zomer-Kooijker (k.zomer-kooijker@umcutrecht.nl)
Francine C van Erp (F.C.vanErp@umcutrecht.nl)
Cornelis K van der Ent (K.vanderEnt@umcutrecht.nl)

**Version:** 2  **Date:** 6 June 2012

**Author's response to reviews:** see over
June 6th, 2012

Dear Editorial Team,

Enclosed you will find our re-submission of the manuscript entitled “The ‘Expert Network and Electronic Portal for children with respiratory and allergic symptoms’: rationale and design”, which we would like to submit for possible evaluation and publication as a ‘Correspondence article’ in BMC Pediatrics.

In reference to your email of April 23rd 2012 we reformatted our manuscript according to the guidelines for Correspondence articles. We have made the following changes:

- **Abstract**
  We have shortened our abstract to a short unstructured, single paragraph summary of 129 words.

- **Main text**
  We added the heading ‘Correspondence’ and changed the heading ‘Results’ in ‘Findings’.

The manuscript has been read and approved by all the authors, and the requirements for authorship have been met. The authors have personally written the complete (100%) manuscript, and this manuscript has not been published elsewhere, nor is submitted elsewhere at this moment.

We will be very grateful if you would consider this manuscript for publication in BMC Pediatrics.

Yours sincerely,

On behalf of all authors,

K. Zomer-Kooijker, M.D.

Email: k.zomer-kooijker@umcutrecht.nl
Telephone number: 0031-88-75 533 53
Fax number: 0031-88-75 547 47